Author's response to reviews

Title: Multidisciplinary care of obese children and adolescents for one year reduces ectopic fat content in liver and skeletal muscle

Authors:

Cilius E Fonvig (crfo@regionsjaelland.dk)
Elizaveta Chabanova (Elizaveta.Chabanova@regionh.dk)
Johanne D Ohrt (jodo@regionsjaelland.dk)
Louise A Nielsen (loani@regionsjaelland.dk)
Oluf Pedersen (oluf@sund.ku.dk)
Torben Hansen (torben.hansen@sund.ku.dk)
Henrik S Thomsen (Henrik.Thomsen@regionh.dk)
Jens-Christian Holm (jhom@regionsjaelland.dk)

Version: 6 Date: 10 November 2015

Author's response to reviews: see over
Att. BMC Pediatrics Editorial Board

Please receive this re-revised manuscript of the original article “Multidisciplinary care of obese children and adolescents for one year reduces ectopic fat content in liver and skeletal muscle”.

Below are the point-by-point response to the editor’s concerns:

Editor’s comment:

"Thank you for adding the STROBE checklist. Could you add page numbers to this indicating where each item is addressed please? Please also add a brief statement to the methods section of the manuscript that the study has been reported in line with the STROBE guidelines."

Page numbers have been added to the uploaded STROBE checklist. In the methods section, the sentence “This study has been reported in line with the STROBE guidelines.” has been added (line 181).

Best regards

Cilius Esmann Fonvig, MD, PhD
The Children’s Obesity Clinic, Department of Pediatrics
Copenhagen University Hospital Holbæk
Smedelundsgade 60, 4300 Holbæk, Denmark
Phone: + 45 28 59 04 56. Fax: + 45 59 48 42 09 E-mail: crfo@regionsjaelland.dk